Massachusetts Institute of Technology Handout 0 6.042J/18.062J: Mathematics for Computer Science September 7, 1995 Professors David Karger and Tom Leighton

Information Sheet

In addition to using this information to determine section assignments, we will distribute a class list with phone numbers and email addresses to facilitate collaboration. If you do not want your phone number or email address distributed by us, please say so below.

Name

Department and year

Phone number

Email address

Please list all Course VI and Course XVIII subjects you have taken or are currently taking. Circle those you are currently taking.

Each student must attend a one-hour tutorial on Monday. Please circle the hours you prefer and cross out those you cannot attend.

9 10 11 12 1 2 3

Recitations will be held on Fridays. Recitations are scheduled for 10:00 (36-372), 11:00 (36-372), 1:00 (26-310), and 3:00 (26-310). Please circle the hours you prefer and cross out those you cannot attend. For tomorrow, September 8, you may attend the recitation of your choice.

10 11 1 3